Author's response to reviews

Title: Alterations in osteoclast function and phenotype induced by different inhibitors of bone resorption - implications for osteoclast quality

Authors:

Anita V Neutzsky-Wulff (avn@nordicbioscience.com)
Mette G Sørensen (mgs@nordicbioscience.com)
Dino Kocijancic (dko@nordicbioscience.com)
Diana J Leeming (djl@nordicbioscience.com)
Morten H Dziegiel (morten.dziegiel@rh.regionh.dk)
Morten A Karsdal (mk@nordicbioscience.com)
Kim Henriksen (kh@nordicbioscience.com)

Version: 4 Date: 27 May 2010

Author's response to reviews: see over
Dear editor: Cintia Cheong and production editor: Philip Dooner.

Thank you for the revisions needed for publication.

The following changes have been made to the manuscript:

1) The style of the author list has been changed to the format: ‘First name Initial Last name’.

2) The title has been changed to sentence case.

3) The list with e-mails has been changed using author initials instead of full names.

4) Information regarding funding has been removed from the title page, and this information is now to be found in the acknowledgement section.

5) The following content has been removed from the title page: Keywords, word-counts and running title.

6) The list of abbreviations has been placed after the conclusion section.

7) The section with the reference list is now entitled ‘References’.

8) Full details regarding the two highlighted references are now given.

9) The figures have been modified so that they no longer contain figure number or headings.

10) Vertical lines have been removed from table 1. (The vertical lines in Figure 6, which has a ‘table-like format’ has not been removed, as this is a figure and not a table. If it is required, we can of course remove these lines).

11) Minor corrections in typography have been made.

Please let us know if further modifications are necessary.

Best regards,
Anita